ECONSTOR Make Your Publications Visible.

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Jackson, Osborne; Kawano, Laura

Working Paper Do increases in subsidized housing reduce the incidence of homelessness? Vidence from the low-income housing tax credit

Working Papers, No. 15-11

Provided in Cooperation with: Federal Reserve Bank of Boston

Suggested Citation: Jackson, Osborne; Kawano, Laura (2015) : Do increases in subsidized housing reduce the incidence of homelessness? Vidence from the low-income housing tax credit, Working Papers, No. 15-11, Federal Reserve Bank of Boston, Boston, MA

This Version is available at: https://hdl.handle.net/10419/130694

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

FEDERAL RESERVE

No. 15-11

Do Increases in Subsidized Housing Reduce the Incidence of Homelessness? Evidence from the Low-Income Housing Tax Credit

Osborne Jackson and Laura Kawano

Abstract:

We examine the impact of subsidized housing on homelessness using the Low-Income Housing Tax Credit (LIHTC), the largest place-based housing program in the United States. To generate quasi-experimental variation in housing placements, we exploit a discontinuous increase in the amount of tax credits available to projects placed in certain high-poverty neighborhoods. Using data from the U.S. Census and HUD, we find that LIHTC project installation has no significant impact on neighborhood homelessness but does significantly reduce county-level homelessness. Our analysis suggests that the mobility of the homeless across neighborhoods helps to explain this result. These findings are consistent with the impacts of place-based policies crossing local boundaries and homeless housing demand that is sensitive to rental prices.

JEL Classifications: H20, H31, I32, R21, R31 Keywords: low-income housing, tax credits, homelessness, regression discontinuity

Osborne Jackson is an economist in the New England Public Policy Center in the research department of the Federal Reserve Bank of Boston. His email address is <u>osborne.jackson@bos.frb.org</u>. Laura Kawano is a financial economist in the Office of Tax Analysis of the U.S. Department of Treasury. Her email address is Laura.Kawano@treasury.gov.

We thank Sebastien Bradley, Greg Burge, Naomi Feldman, Matthew Freedman, Yolanda Kodrzycki, Emily Owens, Todd Pugatch, Bob Triest, and seminar participants at Drexel University, the Federal Reserve Bank of Boston, Northeastern University, the University of Michigan, and the National Tax Association and American Economic Association conferences for their helpful comments. We are grateful to Matthew Freedman and Emily Owens for providing us data and code to compute QCT eligibility for all census tracts. Excellent research assistance was provided by Kathleen Hourihan.

This paper, which may be revised, is available on the web site of the Federal Reserve Bank of Boston at <u>http://www.bostonfed.org/economic/wp/index.htm</u>.

The views expressed in this paper are those of the authors and do not necessarily reflect the policy of the U.S. Department of Treasury, the Federal Reserve Bank of Boston, or the Federal Reserve System. All errors are our own.

This version: May 2015

1 Introduction

The provision of affordable housing for low-income families is often cited by policymakers and advocacy groups as a necessity for ending homelessness (for example, Department of Housing and Urban Development (2011), United States Interagency Council on Homelessness (2010), Urban Institute (2012)).¹ The U.S. government spends a considerable amount on housing programs for the nation's poor,² and the use of federal housing programs to mitigate homelessness has attracted increasing interest following the recent financial downturn and housing market crisis.³ While important for housing policy, however, whether subsidized housing is effective for combating homelessness remains an unresolved question.

The paper addresses this question by estimating the impact on the incidence of homelessness of the Low-Income Housing Tax Credit (LIHTC), the largest and fastest-growing place-based federal program for low-income housing. LIHTC provides a tax incentive for housing developers to create affordable housing by reducing the cost of projects that serve tenants who meet certain income requirements. Using data from the Department of Housing and Urban Development (HUD) and decennial U.S. censuses, we estimate the relationship between local area homeless counts and rental housing units funded by LIHTC. As a source of identifying variation, we exploit an institutional feature of the LIHTC program that generates a discontinuity in the formula that determines the amount of credits available to projects. Although tax credits are generally allocated based on project costs and the percentage of units set aside for low-income households, projects are eligible for additional tax

¹The former president of the National Low Income Housing Coalition, Barry Zigas, stated in his 1988 statement to the Ways and Means Committee of U.S. Congress for a hearing on the Low-Income Housing Tax Credit and the role of tax policy in preserving the stock of low-income housing, "Low income people face an unprecedented housing crisis. A principal cause of homelessness is the severe lack of affordable housing for low income families."

²For instance, in 2011, the Department of Housing and Urban Development (HUD) spent over \$34 billion on rental assistance (tenant-based and place-based), public housing operations and revitalization, and homelessness assistance grants (Department of Housing and Urban Development 2013).

³The Homelessness Prevention and Rapid Re-Housing Program (HPRP, a \$1.5 billion, three-year federal initiative funded through the American Recovery and Reinvestment Act of 2009) was enacted to mitigate homelessness due to the Great Recession, assisting nearly 700,000 individuals in 2010 and likely preventing homelessness rates from increasing even further than observed (Department of Housing and Urban Development 2011).

credits when situated in high-poverty areas known as "qualified census tracts" (QCTs). We use the plausibly exogenous variation in tax credit allocation based on QCT eligibility to identify the effect of increased LIHTC development on the incidence of homelessness.

We find a significantly positive impact of QCT eligibility on the number of LIHTC-funded low-income units installed in a given tract, consistent with previous work (for example, Baum-Snow and Marion (2009)). However, this increase in housing infrastructure for lowincome households has no appreciable impact on neighborhood homelessness, statistically or economically. Because the homeless population may be mobile, we additionally estimate the relationship between subsidized housing and homelessness at the county level and find evidence that increases in LIHTC units reduce county homelessness. Together, these results suggest that LIHTC development attracts homeless individuals to a neighborhood, thus dispersing the impact of a local housing increase more broadly. Such a result would add to a growing theoretical and empirical literature examining the extent to which the effects of place-based policies cross local boundaries (for example, Busso, Gregory, and Kline (2013), Kline and Moretti (2014)).

We also explore factors other than mobility that may affect the impact of local LIHTC development on neighborhood homelessness. First, examining proxies for the housing quality of the housed, we find limited evidence that LIHTC units are allocated in part to individuals previously in lower-quality housing. Second, we examine whether there are spillovers across neighborhoods in the supply of LIHTC housing. We find a negative correlation of low-income housing development across tracts that is quite modest, suggesting that such supply spillovers do not account for the fact that subsidized housing has no significant effect on homelessness at the neighborhood level. Lastly, we develop a supply-demand model to further examine the mechanisms that could drive the non-mobility effect of subsidized housing on homelessness. Our model allows for the influence of various parameters, including the crowd-out of housing that would otherwise have been provided by the private sector (Baum-Snow and Marion (2009); Eriksen and Rosenthal (2010); Malpezzi and Vandell (2002); Sinai and Waldfogel

(2005)), the price elasticity of homeless housing demand, the extent of LIHTC allocation to the homeless rather than the already housed, and the aforementioned spillovers in lowincome housing supply across neighborhoods. Our results imply that the local housing demand of homeless individuals and those on the margin of homelessness may be sensitive to housing prices, with demand elasticities larger than those of the non-homeless in other studies (Hanushek and Quigley 1980).

Our study contributes to the literature that examines, with mixed evidence, the effect of subsidized housing on homelessness. Early (1998) uses homeless count data from a 20-city, 1987 Urban Institute survey matched to data on the housed poor from the American Housing Survey to estimate the impact of city characteristics on the probability of being homeless. He finds that subsidized housing programs are not effective at reducing homelessness, due to a lack of targeting to the homeless. Early and Olsen (2002) use homeless counts from the 1990 Decennial Census and find nonrobust evidence that subsidized housing reduces homelessness rates. Troutman, Jackson, and Ekelund (1999) combine the 1990 Census data with data from a 1984 HUD survey and find positive effects on homelessness of federal spending on low-income housing. They argue that the replacement of affordable housing for the sheltered poor with federally assisted housing creates perverse incentives to become homeless. Moulton (2010) utilizes recent point-in-time estimates of the homeless population produced by HUD and finds that increases in federal homeless funding are correlated with reductions in chronic homelessness.

A related literature examines the effect of housing market conditions, more generally, on the incidence of homelessness. Honig and Filer (1993) find a strong correlation between housing cost measures (rent at the 10th percentile and vacancy rates) and homelessness, using the 1984 HUD survey. Quigley, Raphael, and Smolensky (2001) examine the incidence of homelessness using four data sources, two of which are nationally representative (including 1990 Census data) and two of which are California-based. They find that small increases in affordable housing (higher vacancy rates and lower median rents) may significantly reduce the rate of homelessness. Studies that have highlighted the effect of rent control laws on homelessness find either no statistical effect (Early and Olsen 1998) or a slightly positive effect (Grimes and Chressanthis 1997).

This paper improves upon this earlier work by using more reliable, nationally representative data, by focusing on a large and growing subsidized housing program, and by employing a more credible identification strategy. By utilizing exogenous variation in federal funds through QCT eligibility rather than cross-sectional correlations, we more plausibly identify the causal effect of such funds on the incidence of homelessness. A few studies have used the variation generated by QCT determination to identify the impact of LIHTC on other outcomes like the stock of low-income housing, homeowner turnover rates, and property values (Baum-Snow and Marion 2009), crime (Freedman and Owens 2011), as well as poverty concentration and neighborhood inequality (Freedman and McGavock 2015).

The remainder of the paper is organized as follows: Section 2 provides an overview of the LIHTC program. Section 3 describes the estimation strategy and data used for analysis, while Section 4 presents the main results. Section 5 discusses possible mechanisms for our findings not already explored in the main results, and finally, Section 6 concludes.

2 Overview of the Low-Income Housing Tax Credit

LIHTC is a tax incentive that was created under the Tax Reform Act of 1986 and benefits affordable rental housing developments targeted at low-income households. In general, credits are allocated annually to state housing authorities based on state population and are then distributed to private developers through a competitive process. These credits are nonrefundable and can either be used to offset the developer's tax liability or, as is most often the case, sold to generate capital. Taxpayers who own LIHTCs can claim a dollar-for-dollar reduction in tax liability over 10 years.

To qualify for the credit, a project must meet certain criteria over tenant incomes for a

subset of units. Specifically, either at least 20 percent of the units must be rent-restricted and occupied by individuals whose income is 50 percent or less of area median income (AMI), or else at least 40 percent of the units must be rent-restricted and occupied by individuals whose income is 60 percent or less of AMI, where "area" is defined by the relevant metropolitan statistical area (MSA).⁴

The tax credit amount that a project may receive depends on several factors. A project's qualified basis for the credit is determined by multiplying the "eligible basis," or the costs that are eligible to receive the credit,⁵ by the percentage of units that are allocated to households below the specified income limit. Credit allocations are determined by applying the appropriate credit rate to this qualified basis. There are two credit rates that can apply. The credit rate is equal to a percentage of the cost of development, either up to 70 percent or up to 30 percent of incurred costs in present discounted terms. Projects for new construction and the cost of rehabilitating a building (if not also funded by tax-exempt bonds) are eligible for the 70-percent credit. The cost of acquiring a pre-existing building and projects that are partially funded with tax-exempt bonds are eligible for the 30-percent credit.⁶

While credit allocations generally follow a formula based on project characteristics, there is a deviation allowed from this rule. Under the Omnibus Reconciliation Act of 1989, new legislation provided additional incentives to develop LIHTC projects in certain very-lowincome areas, designated as "qualified census tracts" (QCTs). A tract where at least 50 percent of households have incomes below 60 percent of AMI is eligible to be deemed a QCT.⁷ A project placed in a designated QCT receives an additional 30 percent boost to

⁴Tenant incomes must be initially certified to demonstrate eligibility and are subject annual recertification, with the exception of projects that are 100- to percent LIHTC units (see HUD informational page, "Certifying Tenant Incomes" athttp://www.hud.gov/offices/cpd/affordablehousing/training/web/lihtc/eligibility/incomes.cfm). Additionally, HUD follows the Office of Management and Budget definitions of metropolitan areas, with some exceptions. AMI estimates for all metropolitan areas and counties are made using data from the decennial census, American Community Survey, Internal Revenue Service, and the American Housing Survey.

⁵These are typically the non-land acquisition costs and construction costs of the project.

⁶Annual credit rates are typically around 9 percent or 4 percent; actual percentages are determined monthly by the Internal Revenue Service.

⁷The relevant income threshold for each tract is adjusted according to the average household size in the tract. AMI is decreased by 10 percent for every fewer person than four in the average household, and

its qualified basis. Because of this allocation rule, observationally similar census tracts may face distinct tax credit benefits for LIHTC-funded projects due to differences in QCT status. QCT eligibility is the source of variation we exploit to identify the impact of the LIHTC program on homelessness, using a regression discontinuity framework.

Although the concept of QCTs was created for purposes of the LIHTC program, QCT status is also one of the criteria that can determine whether an area is in a historically underutilized business zone (HUBZone).⁸ The HUBZone Empowerment Contracting Program was enacted into law as part of the Small Business Reauthorization Act of 1997 to encourage economic activity and employment growth in designated HUBZones by providing preferences and access to increased federal contracting opportunities (United States Small Business Administration 2011). We discuss potential concerns for our identification strategy from this additional role of QCT status and how we address such concerns in Section 3.

3 Methodology and data

3.1 Estimation strategy

Our goal is to estimate the impact of the installation of nearby LIHTC units, L, on the number of local area homeless, H. We therefore seek to estimate the following general specification for neighborhood i:

$$H_i = \beta_0 + \beta_1 L_i + \mathbf{X}'_i \theta + \varepsilon_i, \tag{1}$$

increased by 8 percent for every additional person beyond four. Corresponding adjustments are made for noninteger tract average household sizes above or below four. Not all census tracts that meet the requirements for QCT status are designated QCTs because at most 20 percent of a metropolitan area can be designated qualified. When more than 20 percent of the population of a metropolitan area resides in eligible tracts, tracts are ranked according to the fraction of households that meet the income criterion and are designated as QCTs until the 20 percent threshold is met. Baum-Snow and Marion (2009) show that this restriction is not particularly binding, as 96 percent of QCT-eligible tracts in their data are classified as QCTs. The method for distinguishing QCT status has changed since 2000 to incorporate poverty rates as well, but that rule change is not applicable to this study.

⁸In addition to QCT status, an area can be designated as a HUBZone if it is any one of the following: a Qualified Nonmetropolitan County, a Qualified Indian Reservation, a Qualified Base Closure Area, or a Redesignated Area (United States Small Business Administration 2011).

where **X** is a vector of control variables containing other relevant neighborhood characteristics that may affect the number of homeless individuals or LIHTC development. The parameter of interest is β_1 , which indicates the impact of an additional LIHTC unit on an area's homeless count. If low-income housing reduces homelessness, then β_1 should be negative. We specify our dependent variable as the homeless count rather than the rate so that our assumption of linearity and a constant marginal effect β_1 is more likely to hold.⁹

There are several reasons why the number of low-income housing units in an area may be endogenous to the number of homeless. Unobservable neighborhood characteristics or preferences of residents that influence homelessness are likely to be correlated with LIHTC project placement, which is a combination of both having applied for the credit and having received approval of the project application. Moreover, the homeless population itself may factor into LIHTC project placement decisions. Thus, ordinary least squares (OLS) estimates of equation (1) will tend to be biased.

To identify the causal effect of low-income housing on the number of homeless, we exploit the discontinuity in the probability that a project receives additional tax credits through the QCT eligibility of its census tract. Recall that when at least 50 percent of households in a tract have incomes below 60 percent of adjusted AMI, the tract is eligible to be a QCT and is generally designated as such.¹⁰ We therefore employ a fuzzy regression discontinuity (RD) design and instrument for the number of low-income housing units placed in an area, with an indicator for whether the tract is QCT-eligible. The first-stage equation for the number of low-income housing units in an area is given by:

$$L_i = \gamma_0 + \gamma_1 QCT_i + \gamma_2 f(elig_i) + \mathbf{X}'_i \psi + u_i, \qquad (2)$$

where QCT indicates the QCT eligibility of a tract and f(elig) is a function of the running

⁹Nevertheless, baseline population (log) is included in \mathbf{X} , and throughout the paper we discuss the implications of our results for the homelessness rate in the average neighborhood.

¹⁰Because at most 20 percent of a metropolitan area's population can live in qualified tracts, as previously described, the probability of being designated a QCT does not jump discretely from 0 to 1 at the threshold of 0.5.

variable, *elig*, which is the fraction of households in a tract that meet the relevant income requirement. The QCT indicator equals 1 if $elig \ge 0.5$, and 0 otherwise. Meanwhile, u is a mean-zero error term. Given the incentives for LIHTC development in QCTs, we expect γ_1 to be positive.

The covariates in **X** are included to capture heterogeneity across census tracts that may affect homelessness or LIHTC development. The demographic characteristics we include are the 1990 homeless count, the log of median household income, the log of total population, average household size, share of blacks, share of Hispanics, share of females, share married, share of population aged 16 to 24, share of population by education category, and the poverty rate. Housing market characteristics included are the log of median rent and rental vacancy rate. These variables are all measured at the census tract level. Because QCT status is used to determine HUBZones, we also include a tract's unemployment rate as a control variable. Consistent estimation requires that for otherwise similar neighborhoods, the homeless population does not utilize QCT eligibility when determining where to locate.¹¹ Even with our controls, individuals with similar preferences or other unobservable characteristics may choose to locate in the same area, inducing a correlation in the error terms of census tracts in those areas. Therefore, in all estimation unless noted otherwise, we include MSA fixed effects, and we cluster standard errors by MSA to allow for an arbitrary variance-covariance structure within a metropolitan area.¹²

In our RD design, consistent estimation of β_1 comes from a strongly nonzero γ_1 and from the assumption that there are no unobservable characteristics that affect homelessness that also cause nonrandom sorting in census tracts across the QCT eligibility threshold. As Baum-Snow and Marion (2009) discuss, such sorting generally seems unlikely to occur. First, since QCT status was updated in 1993 using the 1990 census, individuals could not know in

¹¹Such homeless location choice would bias our instrumental variables (IV) estimates towards OLS. Due to the LIHTC and HUBZone programs, QCT eligibility (or status) may be indicative of housing and labor market growth and thus affect the location decisions of homeless individuals at the margin of otherwise similar census tracts. We examine this homeless mobility concern in Section 5.

¹²Because some MSAs span states but unobservables may vary at the state level, we treat the portions of the MSA in each state as distinct in these cases.

1990 which tracts would or would not be deemed QCT-eligible within a few percentage points of the 50-percent threshold. Secondly, even if some individuals had perfect information on the composition of the relevant census tracts, the census sampled only one in six households for the long-form survey, thereby inducing substantial sampling variation.

One concern for our tract-level analysis is that the homeless population, which may be mobile, could be attracted to areas with increased LIHTC development. Even in the short run, daily mobility of the homeless can exceed the distance covered by a census tract (Jocoy and Del Casino Jr. 2008), and there is evidence that the homeless are more likely to migrate to neighborhoods with available low-income housing and associated services than to other areas.¹³ Moreover, national statistics suggest that the situation of the overwhelming majority of the homeless is not chronic.¹⁴ The housing demand and related mobility of the homeless whose situation is not chronic may be more responsive than that of the chronic homeless to changes in the affordable housing market. More generally, it has been theorized that the impact of place-based policies like LIHTC is affected by migration, which may offset local effects.¹⁵ Thus, tracts with low levels of LIHTC activity may not be appropriate counterfactual examples for tracts with high levels of LIHTC activity, because of inter-tract migration.

To examine this concern regarding mobility, we examine the relationship between LIHTC development and homelessness in a larger geographic area. We assume that spatial relocation of the homeless on the basis of LIHTC development (or QCT eligibility) may occur across

¹³Jocoy and Del Casino Jr. (2008) find that respondents cite a lack of affordable housing or services as among the top reasons for leaving the city in which they first became homeless. Glaeser, Kahn, and Rappaport (2000) similarly find that access to services like public transportation influences location decisions of the poor. Additionally, this relocation decision could even be initiated or assisted by the local government.

¹⁴In 2011, non-chronic homelessness accounted for 83 percent of total U.S. homelessness (Witte (2012)). During our sample period, HUD defined chronic homelessness as "an unaccompanied homeless individual with a disabling condition who has either been continuously homeless for a year or more or has had at least four episodes of homelessness in the past three years." It was not until the 2009 Homeless Emergency Assistance and Rapid Transition to Housing Act that this definition was amended to include families (Witte (2012)).

¹⁵For a recent discussion and survey of the theory, see Kline and Moretti (2014). Recent examples of empirical work include Freedman and Owens (2011), who discuss potential residential displacement in the context of LIHTC's impact on local area crime, and Busso, Gregory, and Kline (2013), who examine the extent of such relocation when assessing the welfare impact of the Empowerment Zone program.

tracts but not across counties. If this assumption holds, then county-level estimates do not suffer from the location choice issues affecting tract-level analysis. We re-run our analysis at the county level rather than the tract level, assuming that county IV estimates are consistent and that they exclude changes in the homeless count resulting from mobility-driven changes in the composition of a tract (that is, the extent to which it is inhabited by the homeless vs. the non-homeless). Only changes in homelessness due to adjustments in housing status should be observed. Because QCT eligibility is designated at the tract level and not the county level, we use the number of tracts in a county that are QCT-eligible to instrument for county-level LIHTC development.

To ensure consistent county-level IV estimates, analogous to identification at the tract level, the ideal experiment is to compare two equally sized counties with the same number of tracts, N, and with very similar poverty distributions. In particular, these counties would be identical in (N - 1) tracts but differ slightly in the Nth tract, resulting in a difference of one QCT-eligible tract across the two counties.¹⁶ This ensures that the counties would differ in LIHTC development based exogenously on the difference in QCT-eligibility in those otherwise-identical Nth tracts, rather than based endogenously on unobserved differences in poverty that might also affect homelessness. To implement this experimental ideal in countylevel estimation, we include controls for county size via the log of total population and the log of tracts in the county. We also try to control for the distribution of county-level poverty. The poverty controls include a third-degree polynomial analog of the tract-level running variable based on the number of "eligible" households in a county,¹⁷ county-level analogs of all other tract-level controls, and additional higher-order terms for some controls.¹⁸

¹⁶Specifically, assume that in (N-1) tracts across the two counties, the fraction of eligible households is identical, resulting in $J \leq (N-1)$ QCT-eligible tracts in each county. In the last, Nth, tract, both counties again have a similar fraction of eligible households. However, one tract is QCT-ineligible, falling just below the relevant threshold, while the other tract is QCT-eligible, falling just above the threshold. Thus, while one county has J QCT-eligible tracts, the other county has (J + 1) QCT-eligible tracts.

¹⁷Since qualified status is determined at the tract level, there is no true eligibility at the county level. However, the variable remains a valid measure of poverty at any level of geographic aggregation.

¹⁸We experimented with different combinations and functional forms for the higher-order control variables, using fit and first-stage strength as a guide for the preferred specification. Additionally, county analogs of tract variables are defined identically but measured at the level of the county rather than the tract. The

Due to greater variation in area population at the county level, there may be increased heteroskedasticity of the estimation errors relative to our tract-level analysis, which is confirmed by diagnostic testing.¹⁹ To account for this, along with continuing to include clustered standard errors, we weight OLS and IV analysis by 1990 county population. The sample for county estimation includes the same areas (that is, both urban and rural) as tract-level analysis, in order to maintain consistent spatial coverage.

3.2 Description of data

Data for our analysis come from several sources. First, we use homeless counts from the 2000 Decennial U.S. Census, which provides a large and recent source of aggregate, cross-sectional data.²⁰ The Census Bureau conducted its Shelter and Street Night, or "S-Night," efforts to enumerate persons sleeping in a wide range of facilities that serve the homeless as well as persons visible at street locations. Counts of those living in shelters were conducted on March 27, 2000, and counts of those living in designated streets on March 29, 2000. In Section 5, we also use total population and several proxies for the housing quality of the housed population from the 2000 Census.

Data on LIHTC developments come from HUD's LIHTC Database.²¹ The database contains information on projects placed in service between 1987 and 2009. We utilize information on project location, the year the project was placed in service, the number of units,

sole exceptions are the log of household median income and the log of median rent. Due to missing census data at the county level, these variables are the unweighted county means of the tract-level variables, rather than direct county-level measurements.

¹⁹Such increased heteroskedasticity is not necessarily the case (Dickens 1990). As a diagnostic, we compare weighted and unweighted IV estimates at the tract and county levels. At both levels, weighted coefficients are similar to unweighted estimates, falling within 95 percent confidence intervals of the former (for example, for new construction units, the weighted tract-level IV estimate is -62.29). This suggests that our estimation is correctly specified and eliminates motivation for weighting unrelated to heteroskedaticity (Solon, Haider, and Wooldridge 2015). However, while standard errors from weighted estimation, compared with unweighted, are slightly larger at the tract level, they are much smaller at the county level (for example, for new construction units, the unweighted county-level IV standard error is 226.92).

²⁰These data are obtained from from the National Historical Geographic Information System (Minnesota Population Center 2011) and are available at nhgis.org.

²¹The database is available at lihtc.huduser.org.

and the type of project.²² The database also provides information on whether the census tract for the development is a QCT.²³

We restrict our analysis to projects that were placed into service between 1994 and 1999 for two reasons. First, we focus on projects in tracts whose QCT status determinations were based on information in the 1990 Decennial Census, and 1994 was the first full year in which this was the case.²⁴ Second, because we examine homelessness as recorded in the 2000 Census, we assume that projects that were placed into service after 1999 have no impact on our outcome of interest. Between 1994 and 1999, there were 8,660 LIHTC projects placed into service, of which 8,132 projects contain census tract identification.²⁵

To determine each tract's QCT eligibility, we obtained data used by HUD based on the 1990 Census.²⁶ We computed the proportion of households in a tract that fall below the relevant MSA income threshold. We then predicted actual QCT status using additional information on the population and ranking of QCT-eligible tracts in an MSA (see section 2).²⁷

We use additional, baseline tract characteristics that may be determinants of neighborhood LIHTC development or homeless counts from the 1990 Decennial U.S. Census (Min-

 $^{24}\mathrm{QCT}$ status of tracts remains constant over the 1994–1999 period.

 $^{^{22}}$ The database also contains some information on the attributes of the targeted population for the project. One targeted population is the homeless, which represents a very small 3.7 percent of projects in the database with nonmissing target information (or 1.4 percent of all projects). We examine the effect of such targeting for our results later in Section 5.

²³Ideally, we would also examine LIHTC tenant characteristics, such as former homelessness, in further detail. Unfortunately, until recently there was no federal mandate requiring the collection of such tenant data, causing little to be known about the characteristics of LIHTC residents (Horn and O'Regan 2011). Given this restriction, Horn and O'Regan (2011) collect data on a limited sample (Massachusetts, Texas, and Delaware) in order to examine the racial composition of LIHTC tenants. They note that, as of 2009, states are now required to collect and provide some tenant data to HUD.

²⁵There are between 1,263 and 1,468 projects placed into service each year, representing a total of 526,436 units and 486,206 low-income units. Originally, 779 projects are missing census tract information. For these projects, we utilize address information in the HUD database to try to determine the appropriate tract designation. We successfully recover 251 projects in this manner.

²⁶We are grateful to Matthew Freedman and Emily Owens for providing us with these data, as well as code for determining QCT eligibility.

²⁷Because the LIHTC Database contains HUD's QCT designations, we verified that our QCT predictions are the same as those in the database, to be confident we are accurately determining QCT eligibility and status. Our QCT predictions matched HUD's QCT designations for all but one project.

nesota Population Center 2011).²⁸ We collected tract-specific information on household income, household size, population, race and ethnicity, gender, marital status, age, education, unemployment rates, poverty rates, median rent, and rental vacancy rates.²⁹ The 1990 Census also contains counts of the homeless population that we use, although the methodology for enumerating the homeless changed between 1990 and 2000 (Smith and Smith 2001). The homeless count recorded in the 1990 Census is included in **X**, so estimated coefficients measure the impact of covariates on the change in homeless counts relative to 1990 levels, which may also help reduce bias.³⁰ Because of the change in the methodology of counting the homeless, inclusion of 1990 homelessness as a control is preferred over directly transforming our dependent variable to the decadal change in homeless counts.³¹ In Section 5, when proxies for the housing quality of the housed population in 2000 are the dependent variables, we include 1990 measures of these variables as controls instead of 1990 homelessness.

While the LIHTC database contains both 1990 and 2000 census tract identification for a given project based on its address, for tracts where no LIHTC development occurs, we use the Census Bureau's 2000 Census Tract Relationship Files to match 1990 census tracts to 2000 census tracts.³² For the tracts we are able to match across the two censuses and the rest of our sources, we aggregate the combined data up to the census tract level, resulting in 31,573 tract-level observations, of which 30,933 have nonmissing values of all variables and are in our estimation sample with all observations.

²⁸The data are available at nhgis.org.

 $^{^{29}}$ Due to measurement error, in some 1990 Census tracts the proportion of the tract population calculated to be in a particular demographic category was greater than 1. As the number of such observations never exceeded 0.6 percent of all 61,349 1990 Census tract observations that we examined, we censored such values at 1 in our data.

³⁰We thus estimate the impact of the flow in LIHTC projects from 1994 to 1999 on the decadal flow in homeless counts. If LIHTC unit installation is correlated with time-invariant area unobservables, adding 1990 homelessness to the model reduces estimation bias.

 $^{^{31}}$ The transformation imposes the restriction that the coefficient on 1990 homeless counts in equation (1) equals one, which may not hold given the measurement differences across census years.

³²The data are available at census.gov/geo/www/relate/rel_tract.html. Some census tracts change geography over the 1990-2000 period. Because our estimation strategy is based on being able to determine the proportion of low-income households in a tract and QCT eligibility, which in turn are based on 1990 Census information, we restrict our sample to tracts whose geography either exhibited no change or a split. We drop tracts that merged or otherwise revised boundaries over the decade. All tract-level statistics and analysis are calculated based on 1990 tract boundaries.

Table 1 presents summary statistics for our data. Importantly, for most of the included demographic and housing characteristics, there is no discontinuity across the policy threshold. Only the share aged 16 to 24, the unemployment rate, and the share with a high school degree are statistically different at the 5-percent level, and the magnitudes of these differences are quite small. On average, compared with QCT-ineligible neighborhoods, eligible areas are poorer, with larger minority populations, higher unemployment and poverty rates, lower levels of educational attainment, and more LIHTC activity. Our estimation strategy exploits the exogenous variation in LIHTC development that occurs in the vicinity of the QCT eligibility threshold. Thus, we estimate the causal relationship between low-income housing and homelessness for these moderately poor neighborhoods around the eligibility cutoff.

3.3 Measurement Error

Given the inherent difficulties in accurately counting the homeless population, measurement error is a potential concern.³³ To consider the impact on our estimates of measurement error in our dependent variable, suppose that the observed homeless count, H^* , is given by $H_i^* = \lambda H_i + \omega_i$, where H is the true, unobserved homeless count and ω is measurement error. Given equation (1), the observed homeless count can be rewritten as:

$$H_i^* = \beta_0 \lambda + \beta_1 \lambda L_i + \mathbf{X}_i' \theta \lambda + \varepsilon_i^*, \tag{3}$$

where $\varepsilon_i^* = \lambda \varepsilon_i + \omega_i$. If $\lambda = 1$, β_1 is estimated consistently with valid IV estimation but with decreased precision due to larger errors. However, we are particularly concerned that $\lambda < 1$, which would cause attenuation bias to contribute to small estimated effects of LIHTC development on homelessness.

Ideally, we could use a large-scale validation study on the 2000 Census S-Night homeless

³³While classical measurement error would lead to less precise estimates of LIHTC's effect on homelessness, nonclassical measurement error could bias our estimates, with the bias going in either direction.

counts to estimate λ . While no such study exists, we utilize data from a small-scale validation study on the 1990 Census S-Night to inform the nature of this potential bias. The consensus view is that the 1990 S-Night effort notably undercounted the homeless, with the Census Bureau's estimate of 230,000 homeless individuals being approximately one-half the magnitude of other national estimates (Quigley, Raphael, and Smolensky 2001). To assess the quality of the S-Night enumeration, the Census Bureau sponsored research in five cities where "decoys" were positioned at various street locations to determine the extent of such mismeasurement (Martin 1992). The proportion of decoys that were not counted provides a rough estimate of the degree of undercounting across the five cities (Quigley, Raphael, and Smolensky 2001).

The validation study yields a positive correlation of 0.43 between undercounting rates and observed homeless counts across the five metropolitan areas.³⁴ If this pattern holds across all MSAs and is similar for the 2000 S-Night enumeration, then this suggests that $\lambda < 1$. Using the undercount rates to inform us about the magnitude of the measurement error, we obtain an estimate of $\hat{\lambda} = 0.836$, which implies that our estimates of LIHTC's impact on local homelessness are somewhat biased towards zero.³⁵ We obtain an approximate correction for this bias by scaling up our estimates by a factor of $1/\hat{\lambda} = 1.196$, or 19.6 percent. We examine the impact of measurement error on our results as we present them.

³⁴The percentages of decoys not counted was 10 percent in New Orleans, 10 percent in Phoenix, 13 percent in Los Angeles, 20 percent in New York, and 25 percent in Chicago (Martin 1992). Meanwhile, the estimated homeless counts in 1990 based on S-Night were 569 in New Orleans, 1,277 in Phoenix, 5,843 in Los Angeles, 21,986 in New York, and 3,210 in Chicago.

³⁵With a value of $\lambda < 1$, the impact of measurement error on estimate precision is less clear, as this depends on how ε^* compares with ε .

4 Results

4.1 First stage

Before turning to our main specification, we present results from our first stage regressions in Table 2. Each column represents estimates for one of four measures of LIHTC development: the number of LIHTC-funded units, the number of units set aside for low-income households, and these measures focused on new construction projects only. Each cell provides an estimate of γ_1 for a different specification of equation (2). Unless stated otherwise, we include a cubic polynomial of the running variable to allow for sufficient flexibility.³⁶ For our baseline specification, we focus on census tracts where the share of the eligible population is between 25 and 75 percent. In specifications 2–4, we adjust the window of the running variable for tracts included in our estimation sample. In specifications 5–8, we conduct placebo tests where we consider alternative thresholds for determining QCT eligibility to verify the validity of our RD design.

In the baseline specification, the impact of QCT eligibility on LIHTC outcomes for otherwise similar neighborhoods is universally positive, as expected, if additional tax credits lead to more development. The impact of QCT eligibility is both relatively large and stronger for new construction development than for rehabilitation projects, which is consistent with other studies (for example, Baum-Snow and Marion (2009), Freedman and McGavock (2015)).³⁷ A QCT-eligible area receives, for instance, 6.4 more new construction units than an otherwise similar ineligible tract. Given that the average number of new construction units in a tract is 9.7, this constitutes a 66 percent increase. Similarly, QCT eligibility increases new con-

³⁶We experimented with different functions of the running variable, from a linear function to a sixth-degree polynomial. Results are similar across specifications.

³⁷New construction projects account for the majority of LIHTC projects placed into service in our sample. Because new construction projects generally receive a higher credit rate than rehabilitation projects, the impact of an increased basis through QCT status is larger for new construction developments. Another measure of LIHTC development that could be considered is the number of projects that are funded by the credit. The effects of QCT status on the number of projects placed in a tract are not statistically significant, however, indicating that additional benefits of locating a LIHTC project in a QCT do not induce developers to engage in more projects, but rather to expand the size of existing projects by adding units.

struction low-income units by 58 percent (that is, a 5.1 unit increase relative to an average of 8.8).

To better understand what these first-stage estimates imply for the potential impact of LIHTC development on homelessness, note that the average census tract in 2000 had a population of 5,241 and a homelessness rate of 0.7 percent, equivalent to 37 homeless individuals. Under the extreme assumptions that all of the additional 6.4 new construction units received by QCT-eligible tracts are allocated to the homeless, and that the homeless occupy units at the same rate as the non-homeless (that is, 2.62 individuals per unit), then LIHTC development would lead to $6.4 \times 2.62 \approx 17$ fewer homeless people. Thus, our firststage results would imply a 45 percent reduction in the homelessness rate to 0.4 percent. This exercise suggests that there is scope for a large effect of QCT eligibility on homelessness through increases in LIHTC development.³⁸

The first-stage results are illustrated in Figure 1, which plots predicted values of the cubic control functions and average values of LIHTC development variables within percentage point bins against the fraction of eligible households. The left column reflects all LIHTC development, while the right column reflects new construction development only. The remaining specifications in Table 2 confirm the validity of our first-stage estimation. When adjusting the window of the running variable, point estimates are generally consistent with our baseline specification. Additionally, as expected, the discontinuity in LIHTC activity based on placebo eligibility thresholds is typically not statistically different from zero.

4.2 Main specification

Table 3 provides results from equation (1), focusing solely on the two new construction unitbased measures of LIHTC development where credible IV estimation is plausible.³⁹ Columns

 $^{^{38}}$ See Section 5 and the appendix for a more in-depth, refined version of this exercise that reflects a formalized model.

³⁹Stock and Yogo (2005) provide critical values for partial F-statistics on the exclusion restriction to help test for the presence of weak instruments. Critical values for the maximal actual size of a 5-percent Wald test of $\hat{\gamma}_1 = 0$ are 16.38, 8.96, and 6.66 for maximal test sizes of 10, 15, and 20 percent, respectively. The

(1) and (2) provide our IV regression results, where LIHTC development is instrumented with QCT eligibility. The point estimates suggest that the increase in low-income housing associated with QCT-eligibility may reduce homelessness, but these effects are not statistically different from zero. Thus, we are unable to reject the null hypothesis that LIHTC development has no effect on neighborhood-level homelessness. Figure 2 provides further evidence of the reduced-form relationship between homelessness and QCT eligibility. Plotted in the figure are predicted values of the cubic control functions and average values of the number of homeless (as well as homelessness rates) within percentage point bins against the fraction of eligible households. Once again, we observe no evidence of a decrease in homelessness at the eligibility threshold.

If we take the magnitudes of the estimated effects seriously, the impact of LIHTC on homelessness is also not economically significant. For instance, an increase of 100 new construction units, 10 times the amount in the average tract, is estimated to decrease the homeless count by 18.3 individuals. Thus, in order to get a 10-percent reduction in the homeless count in the average tract (2.2 fewer individuals), new construction units would have to more than double in number. Considering this effect in terms of homelessness rates rather than counts yields qualitatively similar conclusions.⁴⁰

Accounting for measurement error does not affect the economic importance of our results. Focusing on Table 3 column (3), our coefficient estimate would change very slightly with the bias correction, from -18.32 to -21.91. Thus, the magnitude of the estimated bias is quite small, mitigating our concern on this issue. Moreover, because the Census Bureau altered

partial F-statistics on QCT eligibility fall between the first two critical values for both new construction units and new construction low-income units. We keep this in mind for inference, as more conservative, weak-instrument robust confidence intervals would generally lead us to be even less likely to reject the null hypothesis that LIHTC development has no significant effect on homelessness.

 $^{^{40}}$ As noted earlier, the average census tract in 2000 has a population of 5,241 and a homelessness rate of 0.7 percent, equivalent to 37 homeless and 5,204 non-homeless. If new construction units doubled, this would decrease the homeless count by 1.8 individuals, equivalent to just a 5-percent reduction in homelessness rates. Moreover, despite potential misspecification of constant marginal effects, this magnitude is similar to what we obtain if we instead estimate equation (1) with the dependent variable as the homelessness rate rather than the count, on a sample with all observations. Such estimation results in an IV coefficient on new construction units of -0.007. While not significant, this nevertheless implies that a doubling of new construction units would reduce homelessness rates by 10 percent in the average tract.

its 2000 S-Night methodology to address some of the 1990 shortcomings, λ is likely closer to one in the 2000 data than the 1990 data. Finally, under certain assumptions, the more relevant estimate of λ , at the eligibility threshold, may be closer to one than our current $\hat{\lambda}$ estimate, which reflects the mean in the validation sample.⁴¹

For comparison to our IV specification, columns (3) and (4) provide estimates from OLS regressions. These specifications show that ignoring the potential endogeneity of LIHTC project placement results in a significantly positive relationship between LIHTC development and homelessness. That said, given the expectedly larger standard errors in IV estimation, the OLS coefficients still fall within most reasonable confidence intervals around our IV estimates. In columns (5)-(8), we focus on the intensive margin effects of LIHTC development on homeless counts. In columns (5) and (6), we examine tracts with at least some homeless individuals. In the homeless count data, there is a mass point at zero, with 33 percent of tracts having no measured homelessness. This large mass indicates that there may be significant nonlinear effects of LIHTC development and other neighborhood characteristics on homelessness that would lead to inconsistent estimation by linear IV. While the estimated coefficients are more negative in these specifications, they are still not statistically significant. This suggests that the linear approximation to the model is fairly reasonable. In columns (7) and (8), we examine the effect of LIHTC development on homelessness for only those tracts with at least some development of the LIHTC measure of interest. Here, the estimated coefficients are smaller in magnitude than those in the main sample specification, and even slightly positive for new construction units. However, these effects remain not statistically significant, in addition to the fact that specifications (5)-(8) all suffer from weak instruments.⁴²

Table 4 presents results from county-level analysis. Compared to tract-level analysis,

⁴¹For instance, if we make the extreme assumption that the decadal increase in homeless counts observed in Table 1 is entirely measurement error, then such error is actually smaller at the threshold than across the overall sample.

⁴²In all but two specifications, we also strongly reject the hypothesis that the coefficient on 1990 homelessness equals one, suggesting that adding this variable as a control is indeed preferable to transforming the dependent variable to the decadal change in homelessness.

the estimated impact of LIHTC development on homeless counts at the county level will be more negative if tract-level migration induces an inflow of homeless individuals. Our result in column (3) is consistent with LIHTC development (and possibly QCT eligibility) attracting the homeless to census tracts within a county. An increase of 100 new construction units is estimated to decrease the homeless count by 457 individuals.⁴³

Taken together with our estimate that a 100 unit increase in new construction decreases tract-level homelessness by 18 individuals, this suggests that the mobility responses of the homeless population could be substantial. This implies that the unresponsiveness of homelessness to LIHTC development at the neighborhood level masks an elastic, housing-driven response that is nullified by a similarly elastic but opposing mobility-driven response. In the appendix, we further discuss using our tract and county estimates to decompose LIHTC's neighborhood-level effect on homelessness into a "housing effect" due to changes in housing status and a "composition effect" due to population mobility.

A large housing effect may be possible, especially since some LIHTC units in our sample have three or four bedrooms and may thus house several individuals. In analysis similar to our first stage, we examine how QCT eligibility affects the size of units developed, finding significant increases in two-, three-, and four-bedroom units, but not in one-bedroom or efficiency units.⁴⁴ There are also several reasons why a large mobility response may be plausible. First, the homeless whose situation is not chronic may be a particularly priceresponsive and mobile subset of the population, especially for within-county moves across census tracts. Second, the degree of mobility partly reflects views by those migrating on

⁴³The estimate in column (4) of Table 4 for new construction low-income units is similarly negative, albeit smaller in magnitude. However, although this coefficient is statistically significant, the estimation suffers from weak instruments, and so we disregard this result. In the case where county-level estimation does not fully purge LIHTC-driven composition effects, in results not shown, we also examine MSA-level estimation. IV estimation of LIHTC's effect on homelessness at the MSA level is very weak and, thus, these results are unreliable.

⁴⁴Ideally, we would also examine information on the average number of previously homeless people and previously housed tenants in LIHTC units of varying sizes. This would help to provide a rough sense of the plausibility of our estimated housing effect. However, although HUD is collecting new data on LIHTC tenants, prior housing experience is not among the characteristics being surveyed. A second-best alternative might be to use information on the average number of residents in LIHTC units targeted and not targeted to the homeless. Such data, however, are not available.

the substitutability of origin and destination areas.⁴⁵ Given our RD design, high-LIHTCactivity and low-LIHTC-activity neighborhoods at the eligibility threshold are extremely close substitutes in terms of area poverty. Lastly, mobility responses to place-based programs can be affected by program features (for example, Kline and Moretti (2014)). Because the LIHTC program does not generally place restrictions on the previous location of new tenants, we might anticipate a larger mobility response than in other place-based programs where such restrictions or incentives exist.⁴⁶

5 Extensions

5.1 Housing Quality of the Housed

In Section 4.2, we examined the role of homeless mobility in the interpretation of tractlevel results. However, there are other factors which could also contribute to the negligible neighborhood-level effect of LIHTC development on homelessness. For instance, LIHTCfunded units may not be allocated to those who are on the margin of homelessness but rather to the moderately poor who are already housed. The rent ceiling on subsidized units is relatively high, at 18 percent of AMI, and tenant income limits are set at fairly high levels with only some incentive to obtain tenants with lower levels of income. While policymakers have long been concerned that the LIHTC program serves only the moderately poor,⁴⁷ recent work examining the tenants of LIHTC units finds evidence that a substantial portion of LIHTC units are being allocated to those below the poverty line (Horn and

 $^{^{45}}$ Busso, Gregory, and Kline (2013) interpret the lack of household migration found in their study as reflecting worker beliefs that Empowerment Zones are poor substitutes for residence in neighborhoods outside of such zones.

⁴⁶For instance, the Empowerment Zone program offered firms wage credits if they employed local residents (Kline and Moretti 2014).

⁴⁷For instance, to encourage the provision of LIHTC-funded units to lower-income households, the Administration's Budget proposals over the last three years have included an option for projects to select income requirements regarding a maximum average tenant income.

O'Regan 2013).⁴⁸ Nevertheless, these poor households may not be those at the margin of homelessness.

To examine this allocation mechanism, we consider the impact of increased LIHTC activity on the housed population in moderately poor neighborhoods. While we are unable to measure the quality of housing directly, several measures available in the census allow us to infer the quality of housing available to households within a neighborhood. We focus on the possibility that LIHTC development may provide access to higher-quality housing in moderately poor areas.⁴⁹ Table 5 examines the impact of LIHTC development on various measures of housing quality of the housed population. The results are suggestive evidence that LIHTC may improve housing quality. For instance, an increase in LIHTC-funded new construction units increases the mean number of rooms per person in units and reduces the median age of structures in the neighborhood. However, neither effect differs significantly from zero. LIHTC development also decreases the share of miscellaneous and one-unit housing structures in a neighborhood. While this may indicate individuals transitioning to higher-quality housing, these effects are likewise not statistically different from zero.⁵⁰

We observe a significant effect of LIHTC development only on the local share of housing structures that are three-to-four units large. Such a result may indicate an increase in housing quality if there are economies of scale in the provision of shared amenities. Meanwhile, the absence of other significant effects of LIHTC activity on housing quality may be due to the data constraining our analysis to the entire housed population in an area rather than just poorer households, who are most likely to be affected.

To further understand our findings, it would be helpful to also know how local rental

 $^{^{48}}$ Horn and O'Regan (2013) find little evidence that LIHTC program participants are bunched along income limits, as only 20 percent of tenants have incomes at or above 50 percent of AMI.

⁴⁹Alternatively, LIHTC may allow housed individuals to obtain access to equal-quality housing at a lower cost. LIHTC tenants may use such a decrease in rental payments to increase their savings or non-housing consumption.

⁵⁰Miscellaneous structures refers to mobile homes, boats, RVs, vans, and other assorted categories. Oneunit structures, if "detached," have open spaces on all four sides or else are joined to only sheds or garages, while if "attached," are joined to another house or building by a dividing wall that goes from ground to roof (Ruggles, Alexander, Genadek, Goeken, Schroeder, and Sobek 2010). Thus, if these small structures are lower quality, LIHTC might allow individuals to move to higher-quality housing.

prices are changing, if at all, in response to LIHTC development. Such development, if it lowers neighborhood rents through an increase in available low-income housing, could affect the homeless population, the housed population, or both. Ideally, we could test this hypothesis by examining the relationship between LIHTC units and rental prices for the very low end of the rental distribution. However, such information is only available in our data for particular points in the rental distribution, namely the 25th, 50th, and 75th percentiles. When we estimate these regressions, the IV relationship between the number of new construction LIHTC units and the rental prices for each of these percentiles is statistically zero. Nevertheless, we remain unable to conclude that these results indicate a null effect of LIHTC on the full distribution of rental prices.

5.2 External Housing Supply Shocks

In addition to potential effects on the housed population, there may be market spillovers besides the mobility of the homeless that also contribute to LIHTC's lack of an effect on tract-level homelessness. For instance, when a LIHTC unit is created, its new residents may come from a low-income housing unit of a neighboring tract, thus creating a housing vacancy in that neighboring area.⁵¹ Alternatively, a developer's decision to construct new LIHTC housing in a local area may affect developments in neighboring areas. Lastly, LIHTC development in a tract may cause migration between neighborhoods, thus changing both the local area composition (as discussed earlier) and the composition of neighborhoods experiencing no LIHTC development.

To allow for this potential spillover effect of housing supply shocks, we adjust estimating

 $^{^{51}}$ More broadly, it may be that occupants of new LIHTC housing vacate units within the same area. However, depending on how such a vacancy then filters through the population (for example, Rosenthal (2014)), it might ultimately generate available housing externally in another neighborhood. In contrast, it could be that occupying LIHTC units induces local residents to occupy additional units across neighborhoods, thus reducing housing vacancies in neighboring areas.

equation (1) to incorporate the installation of "external" LIHTC units, \tilde{L} :

$$H_i = \beta_0 + \beta_1 L_i + \beta_2 \tilde{L}_i + \mathbf{X}'_i \theta + \varepsilon_i, \tag{4}$$

where \tilde{L} represents LIHTC development external to neighborhood *i* but within a county. If $\beta_2 \neq 0$, then $\hat{\beta}_{1,IV_tract}$ is consistently estimated only if LIHTC development across neighborhoods is uncorrelated.

Table 6 provides results from IV estimation of equation (4). We use the number of external tracts in a county that are QCT-eligible to instrument for external LIHTC development. Due to collinearity issues, we include state fixed effects rather than MSA fixed effects as in Table 3. Columns (3) and (4) show that, despite such a change in fixed effects, coefficients remain very similar to the corresponding estimates in columns (3) and (4) of Table 3, although IV estimation is slightly weaker here. With the inclusion of external LIHTC development in columns (1) and (2), we find a negative and statistically significant, albeit small, effect of external development on local area homeless counts.⁵² For instance, an increase of 100 external new construction units, 10 times the number in the average tract, is estimated to decrease the local homeless count by 0.39 individuals. Meanwhile, the effect of internal LIHTC development on homeless counts remains similar to non-spillover analysis, albeit somewhat more negative, and is roughly 50 times larger than the external effect. This suggests that there is only a modest negative correlation of LIHTC development across neighborhoods within a county, resulting in little, if any, bias from the omission of external LIHTC development in earlier estimation. In the appendix, we further describe combining the tract-level estimates here with previous county-level estimation to separately identify internal and external housing and composition effects.

⁵²Given two endogenous variables being instrumented for instead of one, we use the F-statistic form of the Kleibergen and Paap (2006) rk statistic to evaluate the strength of IV estimation. The Stock and Yogo (2005) critical values for the maximal actual size of a 5-percent Wald test of $\hat{\gamma}_1 = \hat{\gamma}_2 = 0$ are 7.03, 4.58, and 3.95 for maximal test sizes of 10, 15, and 20 percent, respectively. The rk statistic on the QCT instruments exceeds the first critical value for both new construction units and new construction low-income units. Thus, we observe that spillover IV estimation is stronger than its non-spillover counterpart.

5.3 A Model of Housing Supply and Demand

In the appendix (Section A.2), we develop a supply and demand model of low-income housing to further characterize the underlying mechanisms driving the housing effect channel on homelessness. We model the supply of low-income housing as comprising both LIHTCfunded projects and other developments that are targeted towards low-income households. On the supply side, the two parameters of interest are the degree of crowd-out between LI-HTC projects and other low-income housing developments, and the extent to which housing developments spill over across neighborhoods. How LIHTC-driven movements in low-income housing supply translate into changes in local homeless counts depends on several factors. In particular, the two parameters of interest here are the price elasticity of demand for low-income housing by the homeless and the proportion of the homeless demanding LIHTC housing who are allocated units.⁵³ We parameterize the model and use it to gain deeper insight into the estimated housing effect of LIHTC on homelessness.

First, we generate an estimate of the maximal potential housing effect of LIHTC development on homeless counts. For each structural parameter, we assume an estimate that would push the potential effect upwards. We show that under these most generous circumstances, the potential impact of 100 newly constructed LIHTC units is a 2,550-person reduction in the homeless count. Our large estimated internal housing market effect accounts for approximately 21 percent of this maximal potential effect.⁵⁴ Evaluated at the 2000 population and LIHTC development levels of the average census tract, our internal housing effect estimate implies that a 10-percent increase in LIHTC local new construction would cause a reduction in the neighborhood homelessness rate of 0.10 percentage points. This is a 14-percent reduc-

⁵³The model does not discuss housing quality. However, if some or all LIHTC housing is of high quality, then there will be demand for LIHTC housing by high-income, non-homeless individuals. Given rental limits, this likely creates excess demand for LIHTC housing, which then enters our model via the proportion of the homeless with access to low-income housing. Alternatively, one could assume that high-quality housing must filter down to the homeless as low-quality housing before it is accessible (Rosenthal 2014). Both approaches allow for a less than one-for-one relationship between high-quality LIHTC housing constructed and low-income housing accessible to the homeless.

⁵⁴We focus on the internal housing effect because, as the appendix shows, the external housing effect is simply a scaled version of the internal effect.

tion in the rate, equivalent to a -1.4 elasticity of homelessness rates with respect to LIHTC development.

We next derive the implied local demand elasticity for housing by the homeless based on the estimated internal housing effect. The more the homeless have access to newly constructed LIHTC units, the more inelastic demand for low-income housing by the homeless must be to rationalize a given housing effect. Across different assumptions regarding access and crowd-out, we typically find price-sensitive local housing demand by the homeless with elasticities as large as -1.66. In contrast, not accounting for possible composition and external effects in estimation results in potentially biased, price-insensitive elasticities ranging from -0.02 to -0.04. Compared with other studies focusing on the non-homeless population (for example, Hanushek and Quigley (1980)), the housing demand elasticities in our study could be somewhat larger because there may be group-specific differences between the homeless and the non-homeless in relevant demand factors. Moreover, because we estimate tract-level housing demand, mobility of the homeless across neighborhoods contributes to such demand being more elastic.

Third, we find evidence of modest low-income housing development spillovers across neighborhoods within a county. A 10-unit increase in neighboring low-income housing is associated with a decrease in local low-income housing of 1.4 units. This negative relationship in low-income housing across neighborhoods may arise from the locational preferences of developers, residents, or both.

Lastly, we find heterogeneous internal housing effects of LIHTC on homelessness. We show that compared with other parameters, the allocation mechanism for housing units has the largest theorized impact on the size of the internal housing effect.

6 Conclusion

In this paper, we find that while the LIHTC program increases the local (tract-level) stock of low-income rental units, there is no evidence that these increases reduce local area homelessness in moderately poor neighborhoods. Our estimated effect of LIHTC development on tract-level homelessness is neither statistically nor economically significant. We find that this negligible effect is not driven by measurement error of the homeless population, effects on the housed population, or spillovers in low-income developments across tracts. Our analysis suggests that increases in low-income housing may attract homeless individuals to areas with greater LIHTC activity. This mobility response of the homeless population may make it difficult to detect a low-income housing effect on homelessness at the local level. Once such mobility across neighborhoods is taken into account via county-level estimation, evidence suggests that LIHTC development does reduce area homelessness. Our study thus reveals that the local housing demand of the homeless may be fairly price sensitive, likely due in part to high mobility across neighborhoods.

These results suggest that the effects on homelessness and possibly other outcomes of local expansions of LIHTC may cross local boundaries. Another important policy implication, stemming from our theoretical model, is that the method of allocating low-income units is a particularly important channel for affecting homelessness. Our empirical analysis of LIHTC's impact on housing quality of the housed provides some additional support for the importance of the way that low-income housing units are allocated. These findings are consistent with proposed policies to encourage income mixing and to target LIHTC housing to those at risk of becoming homeless or those who already are homeless (United States Interagency Council on Homelessness 2010). Further examination of LIHTC unit allocation, including more extensive effects on the already-housed, would be of interest and increasingly feasible with the future availability of LIHTC participant data.









	Percent	of housh	olds eligib	le for LII	HTC				
	0-25%	25-40%	40-45%	45-50%	50-55%	55-60%	60-75%	75-100%	RD Coeff
Homeless count in 2000	16.59	18.59	20.80	23.92	28.77	33.76	42.52	67.42	-0.92
Homeless count in 1990	1.36	1.98	4.43	6.22	8.29	12.77	17.66	33.53	-0.43
Household median income $(\$)$	43019	27168	21850	19809	18451	16910	13506	8769	-210.8^{*}
Population	4263	4044	3866	3754	3835	3670	3325	2595	125.8^{*}
Household average size	2.68	2.58	2.61	2.66	2.70	2.79	2.82	2.67	0.002
Share black	0.06	0.11	0.21	0.29	0.36	0.44	0.54	0.66	0.01
Share Hispanic	0.04	0.06	0.11	0.14	0.17	0.20	0.22	0.14	-0.001
Share female	0.51	0.52	0.52	0.52	0.52	0.52	0.52	0.53	-0.001
Share married	0.48	0.44	0.38	0.34	0.30	0.27	0.23	0.15	-0.004
Share age 16-24	0.12	0.13	0.14	0.15	0.17	0.17	0.17	0.18	0.01^{**}
Share with less than HS degree	0.12	0.21	0.25	0.27	0.28	0.30	0.33	0.35	-0.01^{*}
Share with HS degree	0.21	0.25	0.23	0.22	0.21	0.19	0.18	0.17	0.01^{**}
Unemployment rate	0.04	0.06	0.09	0.11	0.12	0.13	0.17	0.25	-0.01^{***}
Poverty rate	0.02	0.04	0.05	0.06	0.07	0.08	0.09	0.14	-0.001
Median rent $(\$)$	590	409	377	371	380	374	344	265	-1.06
Rental vacancy rate	0.06	0.07	0.08	0.08	0.09	0.09	0.09	0.10	0.003
LIHTC development									
Units $(00s)$	0.16	0.14	0.15	0.16	0.26	0.27	0.24	0.35	0.10^{***}
Low-income (LI) units $(00s)$	0.14	0.13	0.15	0.15	0.24	0.25	0.23	0.33	0.08^{***}
New construction units $(00s)$	0.12	0.08	0.08	0.07	0.15	0.10	0.08	0.10	0.07^{***}
New construction LI units (00s)	0.11	0.08	0.07	0.07	0.13	0.10	0.08	0.09	0.06^{***}
Observations	10036	12272	2331	1639	1137	953	1932	633	
Notes: Authors' calculations based or	n data fror	n the U.S.	Decennial	Censuses a	nd HUD. A	vll census v	rariables are	e from the 19	066
Census except for the 2000 homeless of according to the LIHTC database. The	count. LIF he R.D coel	ff.C develo fficient is th	pment is de ne estimate	etermined I d discontin	by the present of the	ence ot any variable lis	LIHTC ac sted at the	tivity in a tr OCT threshe	act bld.
conditional on a cubic of the running v	variable an	d MSA fixe	d effects. C)nly tracts	with 25-75	percent of e	eligible hous	seholds are u	sed
to compute these KU coefficients.									

Tracts
Census
of
Statistics
Descriptive
÷
Table

OLS
Development (
LIHTC
Eligibility on
QCT
The Impact of
Table 2: $\frac{1}{2}$

		Dep	endent varia	$_{\mathrm{bles}}$	
	All Deve	lopment	New Const.	ruction (NC)	
	\mathbf{Units}	LI units	\mathbf{Units}	LI units	Obs
	(00s)	(00s)	(00s)	(00s)	
(1) Baseline specification	0.091^{***}	0.072^{***}	0.064^{***}	0.051^{***}	20,264
	(0.031)	(0.027)	(0.019)	(0.017)	
F-stat on $QCT = 0$	8.42	7.24	11.20	9.22	
Tracts with alternative eligi	ible fraction	restrictions			
(2) All observations	0.044	0.030	0.046^{**}	0.036^{***}	30,933
	(0.027)	(0.023)	(0.015)	(0.013)	
(3) Between $0.40-0.60$	0.080^{**}	0.062^{*}	0.067^{***}	0.053^{***}	6,060
	(0.039)	(0.035)	(0.022)	(0.020)	
(4) Between $0.45-0.55$	0.093^{**}	0.077^{*}	0.060^{**}	0.048^{**}	2,776
	(0.044)	(0.041)	(0.025)	(0.022)	
Placebo tests					
(5) Eligible fraction ≥ 0.3	0.016	0.010	0.008	0.003	20,264
	(0.019)	(0.017)	(0.013)	(0.012)	
(6) Eligible fraction ≥ 0.4	-0.023	-0.015	-0.026^{*}	-0.020	20,264
	(0.020)	(0.019)	(0.015)	(0.013)	
(7) Eligible fraction ≥ 0.6	-0.094***	-0.084***	-0.026	-0.020	20,264
	(0.033)	(0.031)	(0.021)	(0.020)	
(8) Eligible fraction ≥ 0.7	0.038	0.037	0.017	0.014	20,264
	(0.052)	(0.049)	(0.026)	(0.026)	
* $n < 0.10$. ** $n < 0.05$. *** $n < 0.05$	0.01				

Database and QCT eligibility). All specifications include tract-level controls for the 1990 number share aged 16-24, population share with less than a high school degree, population share with a and 0.6, we replace the third-degree polynomial of the running variable with a linear function of the running variable. When we restrict our sample to tracts with the eligible fraction of households between 0.45 and 0.55, we do not include the running variable. Standard errors clustered at the Notes: Authors' calculations based on data from the U.S. Decennial Census and HUD (LIHTC of homeless, log of household median income, log of total population, population share black, population share Hispanic, population share female, population share married not separated, population high school degree, unemployment rate, poverty rate, log of median rent, rental vacancy rate, and MSA fixed effects. Additionally, most specifications include a third-degree polynomial of the running variable. When we restrict our sample to tracts with the eligible fraction of households between 0.4 MSA level are in parentheses.

Homelessness
on
LIHTC
of
Impact
The
3:
Table

	Depe	ndent vari	iable: 2000	0 Number	of Homele	ess		
	IV, Main	Sample	ō	LS	IV, H	0 < 1	IV, I	0 < 1
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)
NC units (00s)	-18.32		3.89^{*}		-46.65		0.23	
	(38.00)		(2.12)		(59.59)		(53.66)	
NC LI units $(00s)$		-23.10		4.04^{*}		-61.03		-1.90
		(48.29)		(2.16)		(80.15)		(84.68)
1990 No. homeless	0.62^{***}	0.62^{***}	0.62^{***}	0.62^{***}	0.57^{***}	0.57^{***}	0.94^{***}	0.94^{***}
	(0.02)	(0.07)	(0.07)	(0.02)	(0.07)	(0.07)	(0.13)	(0.14)
F-stat on $1990 \text{ H} = 1$	28.22	29.03	31.54	31.22	37.20	39.20	0.23	0.18
F-stat on $QCT = 0$	11.20	9.22			7.78	5.38	1.97	1.41
Observations	20,264	20,264	20,264	20,264	11,757	11,757	2,333	2,325
R^2			0.25	0.25				
* $p < 0.10$, ** $p < 0.05$, **	* $p < 0.01$							

eligibility). The instrument for LIHTC development in IV specifications is QCT eligibility. All specifications include tract-level controls for the 1990 number of homeless, log of household median income, log of total population, population share black, population share Hispanic, population share female, population share married not separated, population share aged 16-24, population share with less than a high school degree, population share with a high school degree, unemployment rate, poverty rate, log of median rent, rental vacancy rate, and MSA fixed effects. IV specifications also Notes: Authors' calculations based on data from the U.S. Decennial Censuses and HUD (LIHTC Database and QCT include a third-degree polynomial of the running variable. Standard errors clustered at the MSA level are in parentheses.

Table 4: The Impact	ot LIH'I'	C on Hom	elessness, C	ounty-level
Dependent va	uriable: 20	dmuN 000	er of Homel	ess
	ML	S	I	/
	(1)	(2)	(3)	(4)
NC units (00s) -8	89.79**		-457.26***	
	(37.43)		(134.47)	
NC LI units (00s)		-5.11		-392.25^{*}
Homolocenoce 1	01***	(25.25)	1 07***	(218.01) 1 $25***$
rate, 1990	(0.09)	(0.07)	(0.27)	(0.15)
F-stat on QCT				
Tracts=0			11.60	0.869
Observations	2,102	2,102	2,102	2,102
R^2	0.99	0.99		
* $p < 0.10, ** p < 0.05, *$	** p < 0.0	1		
Notes: Authors' calculati	ions based	on data fro	m the U.S. De	cennial Cen-
wr Susses and HUD (LIHTC	Database	and QCT e	ligibility). Tl 1000 nonuloi	ie weights in
strument for LIHTC dev	velopment	in IV speci	fications is the	e number of
QCT-eligible tracts in a c	ounty. All	specification	ns include cou	nty-level con-
trols for the 1990 number	er of home	less, log of	household me	dian income,
log of total population, I	log of trac	ts in county +2mme_nem	r and its quad	lratic, popu- uignonia and
higher-order terms, popul	lation share	e female, po	pulation share	married not
separated, population she	are aged 16	-24, popula	tion share wit	h less than a
high school degree and h	igher-orde	r terms, pol	oulation share	with a high
school degree, unemploym	nent rate ar	nd higher-ore	der terms, pov	erty rate and
higher-order terms, log o	f median r	ent and hig	her-order tern	ns, rental va-
cancy rate and higher-ord	er terms, a longlog of	nd MSA fixe	ad effects. IV	specifications
atso include a county-leve ning variable, tract QCT	t anatog or eligibility.	une unira-ae Standard ei	gree polynom rrors chistered	at of the MSA
level are in parentheses.	. (and 0			

Dependent	variables:	Measures	of Housi	ng Qualit	y of the F	Ioused Po	pulation	(2000)
	Hhld		Year	Sha	re of All	Structure	s: No. of	Units
	Size	Rooms	Built	Misc	Η	2	3-4	5-9
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)
NC units $(00s)$	0.015	0.050	3.624	-0.012	-0.048	0.023	0.048^{*}	-0.023
, ,	(0.160)	(0.074)	(3.716)	(0.033)	(0.049)	(0.031)	(0.029)	(0.028)
Observations	20,145	20,153	20,137	20,153	20,153	20,153	20,153	20,153
* $p < 0.10, ** p < 0$	0.05, *** p	< 0.01						
Notes: Authors' c	alculations	based on da	ta from th	e U.S. Dec	ennial Cens	suses and H	IUD (LIHT	\mathbb{C} Database
and QCT eligibilit	ty). The inst	strument fo	r LIHTC d	levelopmen	t in all spe	cifications i	s QCT elig	ibility. The
following are the d	ependent va	ariables in e	ach specific	cation, all n	neasured in	the 2000 C	fensus: (1)	"Hhld Size"
is the mean numb	her of hereo	ne ner houe	(9) (9)	"Boome"	is the mean	n nimher d	of nersons r	ar room in

Table 5: The Impact of LIHTC Development on the Housed (IV)

is the mean number of persons per household; (2) "Rooms" is the mean number of persons per room in units; (3) "Year Built" is the median year that structures in the tract were built; and specifications (4) to (8) each refer to the share of all housing structures in the given category, arranged by number of units in the structure. Specification (4), "Misc" refers to mobile homes, boats, RVs, vans, and other miscellaneous nousing categories. All specifications include a third-degree polynomial of the running variable as well as population, population share black, population share Hispanic, population share female, population share tract-level controls for the relevant dependent variable in 1990, log of household median income, log of total married not separated, population share aged 16-24, population share with less than a high school degree, population share with a high school degree, unemployment rate, poverty rate, log of median rent, rental vacancy rate, and MSA fixed effects. Standard errors clustered at the MSA level are in parentheses.

Dependent variab	le: 2000 N	umber of	Homeless	
	(1)	(2)	(3)	(4)
NC units internal $(00s)$	-17.90		-16.29	
	(40.15)		(39.77)	
NC units external $(00s)$	-0.39**			
~	(0.17)			
NC LI units internal (00s)		-25.07		-21.09
		(53.29)		(51.76)
NC LI units external (00s)		-0.44**		~
		(0.22)		
1990 No. homeless	0.62^{***}	0.62^{***}	0.62^{***}	0.62^{***}
	(0.07)	(0.07)	(0.07)	(0.02)
F-stat on 1990 $H = 1$	29.57	29.82	27.69	28.34
rk-stat on QCT vars	8.249	7.226		
F-stat on $QCT = 0$			9.446	7.360
Observations	20,264	20,264	20,264	$20,\!264$
* $n < 0.10$. ** $n < 0.05$. *** $n < 0.05$	0.01			

Table 6: Spillovers in the Impact of LIHTC on Homelessness (IV)

for internal LIHTC development in IV specifications is the QCT eligibility of a tract, while the instrument for external LIHTC development is the number degree polynomial of the running variable, as well as other tract-level controls 16-24, population share with less than a high school degree, population share Notes: Authors' calculations based on data from the U.S. Decennial Censuses and HUD (LIHTC Database and QCT eligibility). Internal units, as before, refer to LIHTC development within a tract, while external units refer to LItal population, population share black, population share Hispanic, population rental vacancy rate, and state fixed effects. Standard errors clustered at the MSA level are in parentheses. The rk-stat refers to the F-statistic form of the HTC development outside of a tract but within its county. The instrument of QCT-eligible external tracts in a county. All specifications include a thirdfor the 1990 number of homeless, log of household median income, log of toshare female, population share married not separated, population share aged with a high school degree, unemployment rate, poverty rate, log of median rent, Kleibergen and Paap (2006) rk statistic. The null hypothesis is that the test matrix of minimal canonical correlations between the endogenous variables and the instruments is just below full rank (that is, weak), with the alternative being full rank.

References

- Baum-Snow, Nathaniel and Justin Marion. 2009. "The Effects of Low Income Housing Tax Credit Developments on Neighborhoods." *Journal of Public Economics*, 93(5-6): 654–666.
- Busso, Matias, Jesse Gregory, and Patrick Kline. 2013. "Assessing the Incidence and Efficiency of a Prominent Place Based Policy." *American Economic Review*, 103(2): 897–947.
- Department of Housing and Urban Development. 2011. "The 2010 Annual Homeless Assessment Report to Congress."
- Department of Housing and Urban Development. 2013. "FY 2013 Budget."
- Dickens, William. 1990. "Error Components in Grouped Data: Is It Ever Worth Weighting?" *Review of Economics and Statistics*, 72(2): 328–333.
- Early, Dirk. 1998. "The Role of Subsidized Housing in Reducing Homelessness: An Empirical Investigation Using Micro-Data." Journal of Policy Analysis and Management, 17(4): 687–696.
- Early, Dirk and Edgar Olsen. 1998. "Rent Control and Homelessness." Regional Science and Urban Economics, 28(6): 797–816.
- Early, Dirk and Edgar Olsen. 2002. "Subsidized Housing, Emergency Shelters, and Homelessness: An Empirical Investigation Using Data from the 1990 Census." *The B.E. Journal* of Economic Analysis and Policy, 2(1): 1–34.
- Eriksen, Michael and Stuart Rosenthal. 2010. "Crowd Out Effects of Place-Based Subsidized Rental Housing: New Evidence from the LIHTC Program." Journal of Public Economics, 94(11-12): 953–966.
- Freedman, Matthew and Tamara McGavock. 2015. "Low-Income Housing Development, Poverty Concentration, and Neighborhood Inequality." Journal of Policy Analysis and Management, forthcoming.
- Freedman, Matthew and Emily Owens. 2011. "Low-Income Housing Development and Crime." Journal of Urban Economics, 70(2-3): 115–131.
- Glaeser, Edward, Matthew Kahn, and Jordan Rappaport. 2000. "Why Do the Poor Live in Cities?" Working paper, HIER Discussion Paper No. 1891.
- Grimes, Paul and George Chressanthis. 1997. "Assessing the Effect of Rent Control on Homelessness." Journal of Urban Economics, 41(1): 23–37.
- Hanushek, Erik and John Quigley. 1980. "What is the Price Elasticity of Housing Demand?" *Review of Economics and Statistics*, 62(3): 449–454.
- Honig, Marjorie and Randall Filer. 1993. "Causes of Intercity Variation in Homelessness." American Economic Review, 83(1): 248–255.

- Horn, Keren and Katherine O'Regan. 2011. "The Low Income Housing Tax Credit and Racial Segregation." *Housing Policy Debate*, 21(3): 443–473.
- Horn, Keren and Katherine O'Regan. 2013. "What Can We Learn about the Low Income Housing Tax Credit Program by Looking at the Tenants?" *Housing Policy Debate*, 23(3): 597–613.
- Jocoy, Christine and Vincent Del Casino Jr. 2008. "The Mobility of Homeless People and Their Use of Public Transit in Long Beach, California." Working paper, Final Report: METRANS Project no. 06-13.
- Kleibergen, Frank and Richard Paap. 2006. "Generalized Reduced Rank Tests Using the Singular Value Decomposition." *Journal of Econometrics*, 133(1): 97–126.
- Kline, Patrick and Enrico Moretti. 2014. "People, Places and Public Policy: Some Simple Welfare Economics of Local Economic Development Programs." Annual Review of Economics, 6: 629–662.
- Malpezzi, Stephen and Kerry Vandell. 2002. "Does the Low-Income Housing Tax Credit Increase the Supply of Housing?" *Journal of Housing Economics*, 11(4): 360–380.
- Martin, Elizabeth. 1992. "Assessment of S-Night Street Enumeration in the 1990 Census." Evaluation Review, 16: 418–438.
- Minnesota Population Center. 2011. National Historical Geographic Information System: Version 2.0. Minneapolis, MN: University of Minnesota.
- Moulton, Shawn. 2010. "Does Increased Funding for Homeless Programs Reduce Chronic Homelessness?" Southern Economic Journal, 79(3): 600–620.
- Quigley, John, Steven Raphael, and Eugene Smolensky. 2001. "Homeless in America, Homeless in California." *Review of Economics and Statistics*, 83(1): 37–51.
- Rosenthal, Stuart. 2014. "Are Private Markets and Filtering a Viable Source of Low-Income Housing? Estimates from a 'Repeat Income' Model." American Economic Review, 104(2): 687–706.
- Ruggles, Steven, J. Trent Alexander, Katie Genadek, Ronald Goeken, Matthew B. Schroeder, and Matthew Sobek. 2010. *Integrated Public Use Microdata Series: Version 5.0* [Machinereadable database]. Minneapolis: University of Minnesota.
- Sinai, Todd and Joel Waldfogel. 2005. "Do Low-Income Housing Subsidies Increase the Occupied Housing Stock?" Journal of Public Economics, 89(11-12): 2137–2164.
- Smith, Annetta and Denise Smith. 2001. "Emergency and Transitional Shelter Population: 2000." U.S. Census Bureau, Census Special Reports.
- Solon, Gary, Steven Haider, and Jeffrey Wooldridge. 2015. "What Are We Weighting For?" Journal of Human Resources, 50(2): 301–316.

- Stock, James and Motohiro Yogo. 2005. "Testing for Weak Instruments in Linear IV Regression." Identification and Inference for Econometric Models: Essays in Honor of Thomas J. Rothenberg. Eds. J. Stock and D.W.K. Andrews, Cambridge: Cambridge University Press, ch.5.
- Topel, Robert and Sherwin Rosen. 1988. "Housing Investment in the United States." *Journal* of Political Economy, 96(4): 718–740.
- Troutman, William, John Jackson, and Robert Ekelund. 1999. "Public Policy, Perverse Incentives, and the Homeless Problem." *Public Choice*, 98(1-2): 195–212.
- United States Interagency Council on Homelessness. 2010. Opening Doors: Federal Strategic Plan to Prevent and End Homelessness, 2010.
- "Understanding United States Small Business Administration. 2011. the SBA HUBZone Program." Retrieved November 5, 2011,from website: http://www.sba.gov/content/understanding-hubzone-program.
- Urban Institute. 2012. "Preventing Homelessness: Meeting the Challenge." Retrieved July 2, 2012, from Urban Institute website: http://www.urban.org/url.cfm?ID=900476.
- Witte, Peter. 2012. "The State of Homelessness in America 2012: A Research Report on Homelessness." National Alliance to End Homelessness.

A Appendix

A.1 Identifying Housing and Composition Effects

We show how our estimates of the impact of LIHTC development on homelessness at the tract level and county level can be combined to inform the composition and housing effects of the homeless population at the tract level in principle. If LIHTC development causes shifts in the homeless population, these tract-level estimates capture changes in the homeless count through both a change in housing and a change in the population of a tract. We can rewrite the probability limit of $\hat{\beta}_{1,IV,tract}$ as:

$$\text{plim } \hat{\beta}_{1,IV_tract} = \frac{\partial H}{\partial L} = \underbrace{\left(\frac{\partial H}{\partial COMP} \times \frac{\partial COMP}{\partial L}\right)}_{\text{composition effect}} + \underbrace{\left(\frac{\partial H}{\partial LIH} \times \frac{\partial LIH}{\partial L}\right)}_{\text{housing effect}},\tag{5}$$

where, recall, L is nearby LIHTC units and H is the local area number of homeless, while COMP captures the area composition of the homeless and non-homeless populations and LIH is the total supply of low-income housing.⁵⁵ If the composition effect is purged in our county-level estimates, then we can interpret our county-level estimate as the impact of LIHTC on homeless counts due to a change in the housing status of the homeless (that is, plim $\hat{\beta}_{1,IV_county} = \frac{\partial H}{\partial LIH} \times \frac{\partial LIH}{\partial L}$). If we take the point estimates from our empirical analysis of new construction units at

If we take the point estimates from our empirical analysis of new construction units at face value (that is, $\hat{\beta}_{1,IV_tract} = -18.32$ and $\hat{\beta}_{1,IV_county} = -457.26$), this formulation implies that $\frac{\partial H}{\partial L} \approx -18.32 = (438.94) + (-457.26)$. Thus, for every 100 newly constructed LIHTC units, the induced tract-level migration of the homeless (and possibly of the non-homeless) increases the local homeless count by about 439 individuals, while changes in the housing status of the homeless decrease the local homeless count by about 457 individuals.⁵⁶

In our estimation sample in 2000, the average census tract has 10 new construction LIHTC units, a population of 5,241, and a homelessness rate of 0.7 percent, equivalent to approximately 37 homeless and 5,204 non-homeless individuals. New construction units increasing by 10 percent in the average tract would cause a composition effect equivalent to 4.39 homeless individuals migrating to a tract due to new LIHTC development. Meanwhile, this same 10-percent increase in LIHTC new construction would cause a housing effect equivalent to 4.57 homeless individuals becoming housed due to new LIHTC development. In the absence of any composition effects, this would lead to a new homelessness rate of 0.62 percent, a reduction of 0.08 percentage points or 11 percent. This is equivalent to an elasticity of homelessness rates with respect to LIHTC development of -1.1, as opposed to the -0.05 elasticity implied by our baseline results from Section 4.

Next, we allow for the external housing supply shocks that we estimate in section 5. We

 $^{^{55}}$ As discussed, homeless mobility is also a concern, since consistent tract-level IV estimation assumes that, for otherwise similar tracts, the homeless do not use QCT eligibility when choosing where to locate.

⁵⁶The comparable magnitudes of the composition and housing effects are sensible if migration decisions by the homeless are driven by uncertain expectations over the amount of housing likely to be available. Homeless individuals who relocate but do not acquire an affordable housing unit may nevertheless stay because of local services associated with increased low-income housing or the expectation of additional LIHTC development in the future.

can now combine our tract-level and county-level coefficients to approximate the magnitudes of the internal and external composition and housing effects at the tract level. We rewrite the parameters of interest in equation 4 of our spillover analysis to allow for population mobility and housing effects. Specifically:

$$\text{plim } \hat{\beta}_{1,IV_tract} = \underbrace{\left(\frac{\partial H}{\partial COMP} \times \frac{\partial COMP}{\partial L}\right)}_{\text{internal composition effect}} + \underbrace{\left(\frac{\partial H}{\partial LIH} \times \frac{\partial LIH}{\partial L}\right)}_{\text{internal housing effect}}, \tag{6}$$

$$\text{plim } \hat{\beta}_{2,IV_tract} = \underbrace{\left(\frac{\partial H}{\partial COMP} \times \frac{\partial COMP}{\partial \tilde{L}}\right)}_{\text{external composition effect}} + \underbrace{\left(\frac{\partial H}{\partial LIH} \times \frac{\partial LIH}{\partial \tilde{L}}\right)}_{\text{external housing effect}}.$$
(7)

Lastly, assuming that LIHTC-driven migration occurs only between tracts within a county and not between counties, plim $\hat{\beta}_{1,IV_county}$ mentioned earlier is equal to the sum of the internal and external housing effects.

To separately identify the internal and external housing effects, we impose an identifying assumption regarding composition effects. Specifically, we assume that the migration flows of the homeless that drive the composition effects are random. Thus, LIHTC-driven outflows from a tract become evenly dispersed inflows into all other tracts within the county. 57 More specifically:

external composition effect =
$$-\left(\frac{1}{N_{county}-1}\right)\left(\text{internal composition effect}\right),$$
 (8)

where N_{county} is the number of tracts in a given county and can be proxied with $\overline{N_{county}}$, the average number of tracts in counties.⁵⁸ The total composition effect, which is the sum of the internal and external effects, is $\hat{\beta}_{1,IV_tract} + \hat{\beta}_{2,IV_tract} - \hat{\beta}_{1,IV_county}$. Combined with equation (8), we can identify an estimate of the internal composition effect as:

internal composition effect =
$$\left(\frac{\overline{N_{county}} - 1}{\overline{N_{county}} - 2}\right) \left(\hat{\beta}_{1,IV_tract} + \hat{\beta}_{2,IV_tract} - \hat{\beta}_{1,IV_county}\right).$$
 (9)

As equation (9) suggests, there need to be on average more than two tracts per county in

⁵⁷In a two-tract county, this would imply an external composition effect that is of opposite sign and equal magnitude to the internal composition effect. Meanwhile, in a three-tract county, this would result in an external composition effect that is again of opposite sign but one-half the magnitude of the internal composition effect.

⁵⁸Formally, we can rewrite the external composition effect as $\frac{\partial H}{\partial COMP} \times \frac{\partial COMP}{\partial \tilde{L}} \times \frac{\partial COMP}{\partial COMP}$. If the effect of increased LIHTC housing on neighborhood composition does not differ by neighborhood, then $\frac{\partial COMP}{\partial L} =$ $\frac{\partial COMP}{\partial \tilde{L}}$. Thus, the internal and external composition effects differ only by $\frac{\partial COMP}{\partial COMP}$, the spillover effect of a change in external neighborhood composition on internal neighborhood composition. For tract *i*, let $dCOMP_i = -\sum_j dCOMP_j$, where *j* indexes all other tracts in the county. The simplifying assumption of random migration flows means that $dCOMP_j = dCOMP_k \equiv dCOMP \ \forall j \neq k$. This implies for a two-tract county, dCOMP = -dCOMP, so $-1 = \frac{dCOMP}{dCOMP}$. For a three-tract county, $dCOMP = -(dCOMP_1 + dCOMP_2) = -2dCOMP$, so $-\frac{1}{2} = \frac{dCOMP}{dCOMP}$. In general, for a *J*-tract county, $-\frac{1}{(J-1)} = \frac{dCOMP}{dCOMP}$. Equation (8) follows.

order to identify the internal composition effect. In turn, combining equations (8) and (9), the external composition effect is estimated as:

external composition effect =
$$-\left(\frac{1}{\overline{N_{county}}-1}\right)\left(\text{internal composition effect}\right).$$
 (10)

Our estimates of the internal and external composition effects, combined with equations (6) and (7), allow us to separately identify the internal and external housing effects:

internal housing effect =
$$\hat{\beta}_{1,IV_tract}$$
 – internal composition effect, (11)

external housing effect =
$$\hat{\beta}_{2,IV_tract}$$
 – external composition effect. (12)

We find the internal composition effect to be 514.23, the external composition effect to be -75.26, the internal housing effect to be -532.13, and the external housing effect to be 74.87, with all magnitudes corresponding to the impact on homeless counts from an increase of 100 new construction units. The positive external housing effect arises from the negative association between neighboring and local low-income housing mentioned earlier. Due to this negative relationship, we expect that an increase in external housing increases local homelessness due to the decreased availability of local housing.⁵⁹

A.2 A Supply-Demand Model of Low-Income Housing

To understand the mechanisms that determine the impact of LIHTC development on homelessness, we develop a supply and demand model of low-income housing. Variables are as defined in the main text.⁶⁰ There are both direct effects of LIHTC development within a given neighborhood and spillover effects between neighborhoods. The model developed here formalizes the internal and external housing effects contained in equations (6) and (7). We then parameterize the model to provide estimates of the potential magnitudes of these effects as well as other analysis.

Low-income housing is generally supplied through three channels: (1) LIHTC-funded development (LIH_{LIHTC}) , (2) other private-funded development (LIH_{Other}) , and (3) public housing (LIH_{Public}) , such as HUD's public housing programs. The effect of a withinneighborhood increase in LIHTC housing on the overall local supply of low-income housing is therefore given by:

$$\frac{\partial LIH}{\partial L} = \left(\frac{\partial LIH_{Public}}{\partial L} + \frac{\partial LIH_{Other}}{\partial L} + \frac{\partial LIH_{LIHTC}}{\partial L}\right) \equiv \left(\underbrace{\delta_{Pub} + \delta_{Oth}}_{\delta} + 1\right).$$
(13)

 $^{^{59}}$ It is also worth noting that the sum of the internal and external composition effects, 439, and the sum of the internal and external housing effects, -457, are equivalent to the composition and housing effects described by equation 5, estimated without external supply shocks. Thus, the magnitudes of the external effects help to inform the extent of bias in previous estimation of the internal effects. Those prior estimates can thus be reinterpreted as consistent estimates of total effects (that is, internal plus external).

⁶⁰In the model, H represents the local area homelessness, LIH is the local area low-income housing supply, L represents local area LIHTC units, and \tilde{L} represents neighboring LIHTC units.

In equation (13), δ captures how much LIHTC development crowds out other private and public housing developments, and thus determines the sign of $\frac{\partial LIH}{\partial L}$. The supply of local low-income housing may also be affected by LIHTC development

The supply of local low-income housing may also be affected by LIHTC development in neighboring areas through spillovers. This relationship is given by $\frac{\partial LIH}{\partial \tilde{L}}$, which can be rewritten as: $\frac{\partial LIH}{\partial L\tilde{L}H} \times \frac{\partial L\tilde{L}H}{\partial \tilde{L}}$. The second term represents the impact that LIHTC development in a neighborhood has on the supply of low-income housing within that neighborhood. We assume that the degree of crowd-out of other sources of low-income housing that occurs in response to LIHTC development is constant across neighborhoods. Thus, this term is equal to $\delta + 1$ as above. The first term represents the spillovers in low-income housing supply across neighborhoods, which we represent by κ . Put together, the impact of external LIHTC development on internal low-income housing is given by:

$$\frac{\partial LIH}{\partial \tilde{L}} = \kappa(\delta + 1). \tag{14}$$

In both the internal and external housing effects of LIHTC on homelessness, $\frac{\partial H}{\partial LIH}$ represents the effect of an increase in the supply of low-income housing on local area homelessness. This effect operates through those on the margin of homelessness becoming housed in low-income housing units. To understand this component, let us first define $LIH^{D,H}$ as the amount of low-income housing demanded by homeless individuals and LIH^S as the amount of low-income housing supplied. $LIH^{D,H} = D(P,\xi)$ and $LIH^S = S(P,\zeta)$, where P is the price of low-income housing (that is, rent) and ξ and ζ are demand and supply shifters, respectively. In equilibrium, $LIH^{D,H*} = LIH^{S*} = LIH^{H*}$.

The increase in the equilibrium low-income housing stock demanded by the homeless that arises from an exogenous shift in housing supply is as follows:

$$\frac{\partial LIH^{H*}}{\partial \zeta} = \frac{\partial S}{\partial \zeta} \left(\frac{\eta}{\eta - \phi}\right),\tag{15}$$

where $\eta = \frac{\partial \ln LIH^{D,H}}{\partial \ln P} \leq 0$ and $\phi = \frac{\partial \ln LIH^S}{\partial \ln P} \geq 0$ are, respectively, the price elasticities of lowincome housing demand by the homeless and low-income housing supply.⁶¹ We have already defined $\frac{\partial S}{\partial \zeta}$ in the housing effects (equations (13) and (14)). Thus, the only remaining piece that determines $\frac{\partial H}{\partial LIH}$ is $\frac{\eta}{\eta-\phi}$. We write:

$$\frac{\partial H}{\partial LIH} = -\frac{\rho \pi \eta}{\eta - \phi},\tag{16}$$

where π is the proportion of the homeless demanding LIHTC housing at the given rental price with access to such housing, determined by the allocation behavior of low-income housing owners. Meanwhile, ρ is the number of homeless individuals occupying a low-income housing unit.

These components of local low-income housing supply and demand allow us to rewrite

⁶¹Taking total differentials yields $dLIH^{D,H} = \frac{\partial D}{\partial P}dP + \frac{\partial D}{\partial \xi}d\xi$ and $dLIH^{S} = \frac{\partial S}{\partial P}dP + \frac{\partial S}{\partial \zeta}d\zeta$, as well as the equilibrium condition $dLIH^{D,H*} = dLIH^{S*} = dLIH^{H*}$. Setting $d\xi = 0$, solving for $\frac{\partial LIH^{H*}}{\partial \zeta}$, and rewriting in terms of elasticities yields equation (15).

the housing impact of internal LIHTC development on homelessness as:

$$\frac{\partial H}{\partial L} = \frac{\partial H}{\partial LIH} \times \frac{\partial LIH}{\partial L} = -\left(\frac{\rho\pi\eta}{\eta-\phi}\right)\left(\delta+1\right),\tag{17}$$

and rewrite the housing impact of external LIHTC development on homelessness as:

$$\frac{\partial H}{\partial \tilde{L}} = \frac{\partial H}{\partial LIH} \times \frac{\partial LIH}{\partial \tilde{L}} = -\left(\frac{\rho \pi \eta}{\eta - \phi}\right) \left[\kappa \left(\delta + 1\right)\right]. \tag{18}$$

A.3 Computing the Maximal Potential Effect of LIHTC

To quantify the potential responses of the homeless to the placement of LIHTC units, we parameterize δ , η , and π .⁶² Because the internal and external housing effects are identical other than κ , we restrict our attention to the internal effect for this analysis, noting that the implications for the external effect are the same but are scaled by $\kappa \in (-\infty, \infty)$.

To start, we must bound our parameters of interest. Note that $\eta = \frac{\partial \ln LIH^{D,H}}{\partial \ln P} \leq 0$ and $\phi = \frac{\partial \ln LIH^S}{\partial \ln P} \geq 0$ results in $\frac{\eta}{\eta - \phi} \in [0, 1]$.⁶³ We assume that neither crowd-in nor greater than one-for-one crowd-out are possible, which restricts $\delta \in [-1, 0]$ and $(\delta + 1) \in [0, 1]$. Note also that $\pi \in [0, 1]$ and $\rho \geq 0$. Finally, our model allows us to determine a value for $\kappa \in (-\infty, \infty)$, the parameter that represents $\frac{\partial LIH}{\partial L\tilde{I}H}$. We estimate κ with the ratio of external to internal housing effects, (74.9/-532.1) = -0.14.

For an exercise on the *maximum* reasonable internal housing effect of LIHTC on homelessness using the model, we utilize parameter values that push upwards our estimate of the potential impact of LIHTC. For every new unit made available, we assume that the number of homeless that could occupy a unit is 26.7, 10 times the average household size in our sample, because the homeless population may live in more crowded spaces. Because our housing development measures represent every 100 LIHTC units, for this exercise we additionally multiply the number of homeless per unit by 100 units. We assume that the price elasticity of low-income housing supply is 0.3 (one-tenth of the long-run estimate in Topel and Rosen (1988)) and that the price elasticity of homeless low-income housing demand is -6.4, 10 times the long-run estimate in the Hanushek and Quigley (1980) model using data on Pittsburgh. Lastly, we assume that there is no crowd-out and that all the homeless demanding low-income units at equilibrium rental prices have access. Results from this exercise are presented in Section 5.⁶⁴

⁶²The focus on these parameters might be reasonable, for instance, if $d\phi = d\rho = d\kappa = 0$ (for example, these parameters are relatively constant or cannot be changed exogenously).

⁶³The sole exception to this is when $\eta = \phi = 0$, in which case $\frac{\eta}{\eta - \phi}$ is indeterminate.

⁶⁴Note that with a feasible but less reasonable change in assumptions to perfectly inelastic low-income housing supply and perfectly elastic low-income housing demand, the maximum effect would be a 2,670-person reduction in the homeless count for every 100 low-income units built, rather than our stated 2,550-person reduction. Thus, the true upper bound of the internal housing effect is determined by the number of homeless per unit, as alluded to in Section 4 and our discussion of first-stage magnitudes.

A.4 Relative Importance of Mechanisms

Our analysis suggests that δ (crowd-out), π (allocation), and η (homeless housing demand) are dimensions of heterogeneity where one could anticipate differential effects of LIHTC development on homelessness. Our earlier parameterization suggests that $T = \frac{\partial H}{\partial L} \leq 0$. To determine the relative importance of each of the mechanisms that we have identified, we take second derivatives to understand the nature of each mechanism's impact on homelessness, given by:

$$\frac{\partial T}{\partial \delta} = -\left(\frac{\rho \pi \eta}{\eta - \phi}\right) \le 0,\tag{19}$$

$$\frac{\partial T}{\partial \pi} = -\left(\frac{\rho\eta}{\eta - \phi}\right) \left(\delta + 1\right) \le 0,\tag{20}$$

and

$$\frac{\partial T}{\partial \eta} = -\left[\left(\frac{\rho\pi}{\eta-\phi}\right)\left(\delta+1\right)\right]\left[1-\frac{\eta}{\eta-\phi}\right] \ge 0.$$
(21)

Equation (19) shows that reducing the degree to which LIHTC development crowds out other low-income housing developments (that is, by increasing δ) will result in a (weakly) more negative housing effect of LIHTC on homelessness. Similarly, equations (20) and (21), respectively, imply that increasing the proportion of the homeless with access to LIHTC housing units (by increasing π) or increasing the magnitude of the price elasticity of lowincome housing demand by the homeless (by lowering η) results in a (weakly) more negative housing effect of LIHTC on homelessness.

We can compare magnitudes of the marginal effects in equations (19)–(21) to determine each mechanism's importance. In this exercise (available upon request), we use parameter estimates found in the existing literature as a baseline when available.⁶⁵ The exercise shows that the LIHTC housing effect is most sensitive to the allocation mechanism, followed by the degree of development crowd-out, and finally the price-elasticity of housing demand by the homeless (that is, $|\frac{\partial T}{\partial \pi}| \geq |\frac{\partial T}{\partial \theta}| \geq |\frac{\partial T}{\partial \eta}|$).⁶⁶

In additional empirical analysis (available upon request), we use proxies for these parameters to examine directly whether the model's hypotheses hold, interacting the proxies with LIHTC development. The results we obtain are generally consistent with the model's predictions, although not statistically significant.⁶⁷

⁶⁵Specifically, $\delta = -0.2$, implying $\delta + 1 = 0.8$ (Baum-Snow and Marion 2009).

⁶⁶However, this is distinct from cost-effectiveness since, for instance, $d\delta$ may be less costly than $d\pi$, and since homeless count reductions would need to be valued.

⁶⁷We use the log of the median rent as the crowd-out proxy, since as the degree of crowd-out increases, we would expect market rents to remain higher, given the smaller increase in housing supply. We use the share of LIHTC units targeted to the homeless to proxy for the share of the homeless with access to LIHTC units. Finally, since mental illness, drug use, and veteran status may be positive, nonprice determinants of homelessness, such individuals should be more price inelastic in their demand for low-income housing. We therefore use the population shares of these groups as our proxy for demand elasticity.